Author's response to reviews

Title: Quantifying late gadolinium enhancement on CMR provides additional prognostic information in early risk-stratification of nonischemic cardiomyopathy: a cohort study

Authors:

   Pauli Pöyhönen (pauli.poyhonen@helsinki.fi)
   Sari Kivistö (sari.kivisto@hus.fi)
   Miia Holmström (miia.holmstrom@hus.fi)
   Helena Hänninen (helena.hanninen@hus.fi)

Version: 2 Date: 29 March 2014

Author's response to reviews: see over
Submission date: 29.03.2014

Journal: Journal of BMC Cardiovascular Disorders

Dear Editors,

We have made the requested changes to our main manuscript text:

- Addition of line numbering
- Addition of page numbering
- Deletion of page breaks.

Yours Sincerely

Corresponding author:
Pauli Pöyhönen MD, M.Sc.
Division of Cardiology, Heart and Lung Center, Helsinki University Central Hospital
Address: Haartmaninkatu 4, PL 340, 00029 HUS, Helsinki, Finland.
Tel.: +358405464445.
E-mail: pauli.poyhonen@helsinki.fi.